Specific Comments to Authors: The manuscript demonstrated a novel method for determining the accurate size of acetabular cup in the process of primary total hip arthroplasty, its validity and practicable value in normal acetabular reaming and cup implanting. The manuscript is well written, but the limitation should also be mentioned.

Dear Reviewer,

A paragraph describing each possible limitation was added at the end of the results section. Thank you very much for the time spent to review our work, as well as for your recognition.

Kind regards,

Michail Vavourakis